Author’s response to reviews

Title: Compensation of subjects for participation in biomedical research in resource – limited settings: a discussion of practices in Malawi

Authors:

Wongani Nyangulu (wnyangulu@gmail.com)

Randy Mungwira (rgmungwira@gmail.com)

Nginanche Nampota (nnampota@gmail.com)

Lufina Tsirizani (lufinatsiry@gmail.com)

Osward Nyirenda (omnyirenda@gmail.com)

Edson Mwinjiwa (edsonmwinjiwa@gmail.com)

Titus Divala (titus.divala@lshtm.ac.uk)

Version: 2 Date: 31 Oct 2019

Author’s response to reviews:

1. Please be sure to include the following headings in the main text of your manuscript:

   Background
   Main Text
   Conclusions

   AUTHOR RESPONSE - We have added these main headings in the manuscript (Background on page 2 of 12; Main Text on page 4 of 12; Conclusions of page 8 of 12).

   For more information on what should be included in each section, please visit the following page: https://bmcmedethics.biomedcentral.com/submission-guidelines/preparing-your-manuscript/debate

2. Please remove the response to reviewers, as it is no longer needed at this stage.

   AUTHOR RESPONSE - This has been removed accordingly.

3. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.
AUTHOR RESPONSE - I have uploaded the revised submission in the final form for publication. It is a clean copy and all responses are in the input box.

We noted that there were no further actions required from the peer reviewers. Thank you for accepting this manuscript and we look forward to reading it in published form.